\documentclass{sbthesis}
\usepackage{graphicx}
\title{Interesting Title}
\author{Some Student}
\Department
\EmailAddress{student@csci.csusb.edu}
\Advisor{Keith Evan Schubert}
\Committee*{Ernesto Gomez}{Dick Botting}{Casey Jones}
\CSUSBDate{June 2006}
\Copyright
\AbstractText{Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  Some really long winded summary of the thesis.

Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  Some really long winded summary of the thesis.

Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  Some really long winded summary of the thesis.  }
\AcknowledgementText{I would like to thank advisor, the faculty, my mom, dad, my loving family, and most of all God.  I would like to thank advisor, the faculty, my mom, dad, my loving family, and most of all God.  I would like to thank advisor, the faculty, my mom, dad, my loving family, and most of all God.

I would like to thank advisor, the faculty, my mom, dad, my loving family, and most of all God.  I would like to thank advisor, the faculty, my mom, dad, my loving family, and most of all God.

I would like to thank advisor, the faculty, my mom, dad, my loving family, and most of all God.  I would like to thank advisor, the faculty, my mom, dad, my loving family, and most of all God.  I would like to thank advisor, the faculty, my mom, dad, my loving family, and most of all God.  I would like to thank advisor, the faculty, my mom, dad, my loving family, and most of all God.
}
\DedicationText{To Zel and Septimus.}
%\NoTables
\begin{document}
\Thesis
\Chapter{Introduction}
Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.
\section{Background}
Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah\footnote{Blah blah blah.}.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.

Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.

\subsection{A third level}
Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.

\subsubsection{A fourth level}
Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.
\section{Significance}
Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah $x^2$ blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah, see Eq.~\ref{eq-important}.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.

\begin{eqnarray}
f(x) &=& \left(\frac{\sum_{i=1}^n(1+x_i^2)}{(1-x)x^{\pi}}\right)\sin(x)^2 \nonumber \\
     &=& g(x)h(x) \label{eq-important}
\end{eqnarray}


Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.

Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah. For more information see Chapter~\ref{c-scope}, which starts on page~\pageref{c-scope}.
\section{Purpose}
Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah, see Figure~\ref{fig1}.

\begin{figure}
  % Requires \usepackage{graphicx}
  \begin{center}
  \includegraphics[scale=.5]{6view.eps}\end{center}

  \caption{My Figure.}\label{fig1}
\end{figure}


Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah, see \cite{Keith01}.
Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah. For more information see \cite{cggs,cgs99}.

\begin{table}
\caption{Here is my table.}\label{t-table1}
\begin{center}
\begin{tabular}{|lcr|p{1.2in}|}\hline
data 1 & time & something & Comments \\\hline\hline
13214 & 24321 & 3413245 & textual comment textual comment textual comment textual comment textual comment\\
13 & 1 & 34 & textual comment textual comment textual comment textual comment textual comment\\\hline
\end{tabular}
\end{center}

\end{table}

Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.  Blah blah blah blah blah blah blah blah blah blah blah blah blah blah blah.
\Chapter{Something New}
\Chapter{Project Scope}\label{c-scope}
\Chapter{Deliverables}
\appendix
\Appendix{Detailed Proof}

Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this.

\section{More Stuff}
Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this.
\begin{eqnarray}
a^2+b^2 &=& c^2
\end{eqnarray}
Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this. Boy I hope no one reads this.



\begin{figure}
  % Requires \usepackage{graphicx}
  \begin{center}
  \includegraphics[scale=.5]{6view.eps}\end{center}

  \caption{My Figure.}\label{fig2}
\end{figure}

\Appendix{My Code}
Code to go here

\nocite{*}
\Bibliography{keithComp}
\end{document} 